Author's response to reviews

Title: An unusual anatomical variation of superior turbinate: case report

Authors:

Turhan San (turhansan@gmail.com)
Emre Gürkan (dremre_grkn@yahoo.com)
Barış Erdoğan (bariserdogan1981@yahoo.com)

Version: 5 Date: 4 February 2014

Author's response to reviews:

An unusual anatomical variation of superior turbinate: case report.

Figure 2 Axial paranasal CT slice showing anatomical variation of superior turbinate.

Rhinologic causes of headache include deviated nasal septum, congestion of the turbinates, nasal neoplasm, pneumatized agger nasi cells, unusual deflections of uncinated process, paradoxically bent middle turbinate and variations of ethmoid bulla [1]. Also, pneumatized turbinates have been reported as rare cause of headache [2, 3].